Author's response to reviews

Title: An economic appraisal of the Australian Medical Sheepskin for the prevention of sacral pressure ulcers from a nursing home perspective

Authors:

Patriek Mistiaen (p.mistiaen@nivel.nl)
Andre Ament (A.Ament@BEOZ.unimaas.nl)
Anneke L Francke (a.francke@nivel.nl)
Wilco Achterberg (wp.achterberg@vumc.nl)
Ruud Halfens (R.Halfens@ZW.unimaas.nl)
Janneke Huizinga (j.huizinga@derm.umcg.nl)
Henri Post (hpost@evean.nl)

Version: 3 Date: 4 August 2010

Author's response to reviews:

Dear editor,

we have made the final changes to the manuscript as requested. Author affiliations are written fully; A paragraph with authors' contributions has been added; placement ext of tables has been removed and typography has been checked.

We hope the manuscript now fulfills all requirements

sincerely

dr. Patriek Mistiaen